Author’s response to reviews

Title: Simulation and minimization: Technical advances for factorial experiments designed to optimize clinical interventions

Authors:

Jocelyn Kuhn (jocelyn.kuhn@bmc.org)
Radley Sheldrick (rshldrck@bu.edu)
Sarabeth Broder-Fingert (Sarabeth.Broder-Fingert@bmc.org)
Andrea Chu (Andrea.Chu@bmc.org)
Lisa Fortuna (Lisa.Fortuna@bmc.org)
Megan Jordan (Megan.Jordan@dothousehealth.org)
Dana Rubin (Dana.Rubin@bmc.org)
Emily Feinberg (emfeinbe@bu.edu)

Version: 3 Date: 02 Dec 2019

Author’s response to reviews:

1 - Please clarify whether the informed consent was written or verbal for the study and include an amended statement in ‘Ethics approval and consent to participate’ section. If the consent were verbal please give a reason for this and whether the Ethics Board approved it.

Written informed consent was obtained. This clarification was added to page 19.

2 - Please provide specific Authors' contributions using unique initials for all of the authors in the Authors' contributions section.

Initials of all authors were added to the contributions section on page 20.

3 - At this stage, please upload your proofread manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethrough or text in different colours. All relevant tables and figures should also be clean versions. Figures (and additional files) should remain uploaded as separate files. Should you wish to respond to these revision requests, please include the information in the designated input box only.
A final, proofread version has been uploaded accordingly.